\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{512.572} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{20M07} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\newtheorem{example}{Example}


\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
     {
  \vbox{
     \noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
     {\LARGE \textsf{ÑÈÁÈÐÑÊÈÅ ÝËÅÊÒÐÎÍÍÛÅ}} \\[2mm]
     {\LARGE \textsf{ÌÀÒÅÌÀÒÈ×ÅÑÊÈÅ ÈÇÂÅÑÒÈß}} \\[2mm]
     {\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
     {\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
     \raisebox{1ex}{ISSN 1813-3304 }
     \vspace*{-3mm}
%     {\small Èíñòèòóò ìàòåìàòèêè èì. Ñ.Ë.Ñîáîëåâà ÑÎ ÐÀÍ}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
}
     \vspace{-1mm}
     \noindent
      \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
     \hline\hline
     \end{tabular}

     \vspace{3mm}
     {\flushleft\it Vol. 20, No. 2, pp. 144--144 (2023) \hspace{\fill}{\rm\small ÓÄÊ \udcs \hspace{-4mm}}}
     \newline  %Volume, pages in Russian, filled by Editorial board
     	{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \ \mscs }
  }%\vbox
}


\newcommand{\Zbb}{{\mathbb{Z}}}
\newcommand{\R}{{\mathbb{R}}}
\newcommand{\Q}{{\mathbb{Q}}}
\newcommand{\Nbb}{{\mathbb{N}}}

\newcommand{\Ax}{{\mathrm{Ax}}}

\newcommand{\LL}{{\mathcal{L}}}
\newcommand{\Hom}{{\mathrm{Hom}}}
\newcommand{\Res}{{\mathrm{Res}}}
\newcommand{\Dis}{{\mathrm{Dis}}}
\newcommand{\qvar}{{\mathrm{qvar}}}
\newcommand{\pvar}{{\mathrm{pvar}}}
\newcommand{\ucl}{{\mathrm{ucl}}}
\newcommand{\Rad}{{\mathrm{Rad}}}
\newcommand{\Red}{{\mathrm{Red}}}
\newcommand{\var}{{\mathrm{var}}}

\newcommand{\Ss}{{\mathbf{S}}}
\newcommand{\tSs}{{\tilde{\Ss}}}
\newcommand{\tx}{{\tilde{x}}}

\newcommand{\V}{{\mathrm{V}}}
\newcommand{\qq}{{\mathrm{q}_\omega}}
\newcommand{\uu}{{\mathrm{u}_\omega}}

\newcommand{\A}{{\mathcal{A}}}
\newcommand{\T}{{\mathcal{T}}}
\newcommand{\F}{{\mathcal{F}}}
\newcommand{\M}{{\mathcal{M}}}
\renewcommand{\P}{{\mathbf{P}}}
%\newcommand{\C}{{\mathcal{C}}}
\newcommand{\B}{{\mathcal{B}}}
\newcommand{\I}{{\mathcal{I}}}
\newcommand{\D}{{\mathcal{D}}}
\newcommand{\N}{{\mathbb{N}}}
\newcommand{\Pp}{\mathcal{P}}
\newcommand{\Cc}{{\mathcal{C}}}
\newcommand{\Z}{{\mathcal{Z}}}

\newcommand{\lb}{{\langle}}
\newcommand{\rb}{{\rangle}}
\newcommand{\ccl}{{\langle\Ss\rangle}}
\newcommand{\Sing}{{\mathrm{Sing}}}

\renewcommand{\a}{{\mathbf{a}}}
\renewcommand{\b}{{\mathbf{b}}}
\newcommand{\e}{{\mathbf{e}}}
\renewcommand{\d}{{\mathbf{d}}}
\renewcommand{\c}{{\mathbf{c}}}
%\newcommand{\f}{{\mathbf{f}}}
\newcommand{\x}{{\mathbf{x}}}
\newcommand{\y}{{\mathbf{y}}}
\newcommand{\n}{{\mathbf{n}}}

\newcommand{\Nbf}{{\mathbf{N}}}
\newcommand{\Qbf}{{\mathbf{Q}}}
\newcommand{\Ubf}{{\mathbf{U}}}
\newcommand{\Vbf}{{\mathbf{V}}}
\newcommand{\Fbf}{{\mathbf{F}}}
\newcommand{\Zcal}{{\mathcal{Z}}}
\newcommand{\Acal}{{\mathcal{A}}}
\newcommand{\Bcal}{{\mathcal{B}}}
\newcommand{\Ccal}{{\mathcal{C}}}
\newcommand{\Tcal}{{\mathcal{T}}}


\newcommand{\eq}{{\mathbf{eq}}}
\newcommand{\zero}{{\mathbf{0}}}
\newcommand{\one}{{\mathbf{1}}}
\newcommand{\E}{{\mathbf{E}}}
\newcommand{\1}{{^{-1}}}
\newcommand{\eps}{{\varepsilon}}
\newcommand{\inv}{{^{-1}}}

\newcommand{\al}{{\alpha}}
\renewcommand{\t}{{\tau}}
\newcommand{\s}{{\sigma}}
\newcommand{\om}{{\omega}}

\newcommand{\pr}{{\prime}}


\newcommand{\Var}{{\mathrm{Var}}}

\begin{document}
\begin{thebibliography}{1} % Ññûëêè íà ëèòåðàòóðó ïðèâîäÿòñÿ òîëüêî íà àíãëèéñêîì ÿçûêå (äàæå åñëè ó èñòî÷íèêà íåò àíãëîÿçû÷íîé âåðñèè).


\bibitem{1} A. H. Clifford, G. B. Preston
{}{\it The Algebraic Theory of Semigroups: Part I}, AMS, New York, 1961.

\bibitem{2}
E. Daniyarova, A. Myasnikov, V. Remeslennikov,
\href{https://doi.org/10.1007/s10958-012-0923-z}{\it Algebraic geometry over algebraic structures, II: Fundations}, J. Math. Sci., \textbf{185}:3 (2012), 389--416.

\bibitem{3}
T.Harju, J.Karhumaki, W.Plandowski, 
\href{https://doi.org/10.1007/3-540-60084-1_95}{\it Compactness of systems of equations in
semigroups}, International Colloquium on Automata, Languages and Programming,
(ICALP 1995, Szeged, Hungary) LNCS \textbf{944}, (1995), 444--454.

\bibitem{4}
T. Harju, J. Karhumaki, M. Petrich, 
\href{https://doi.org/10.1007/3-540-63246-8_16}{\it Compactness of systems of equations on completely regular semigroups}, Structures in Logic and Computer Science, LNCS \textbf{1261}, (1997), 268--280.

\bibitem{5}
B. I. Plotkin, 
\href{https://doi.org/10.1007/s10958-006-0390-5}{\it Problems in algebra inspired by universal algebraic geometry}, Fundam. Prikl. Mat., \textbf{10}:3, (2004), 181--197.

%\bibitem{rasin}
%V.V. Rasin, 
%\href{https://doi.org/10.1007/BF02194314}{\it On the lattice of varieties of completely simple semigroups}, Semigroup forum, \textbf{17}, (1979), 113--122.

\end{thebibliography}

\renewcommand{\refname}{References}
\renewcommand{\proofname}{Proof.}
\renewcommand{\figurename}{Fig.}




\thispagestyle{empty}

\title[Varieties of semigroups and B.Plotkin`s problem]{\large Equationally Noetherian varieties of semigroups and B.Plotkin`s problem}


\author{{\bf \large \textsc{A.N.~Shevlyakov}}\protect\orcidicon{0000-0002-5338-6264}}



\address{Artem Nikolaevich Shevlyakov  % Êîíòàêòíûå äàííûå âñåõ àâòîðîâ è ìåñòî ðàáîòû óêàçûâàþòñÿ òîëüêî íà àíãëèéñêîì ÿçûêå
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} Pevtsova st. 13,
\newline\hphantom{iii} 644099, Omsk, Russia}%
\email{\textcolor{blue}{a\_shevl@mail.ru}}%

\thanks{\sc Shevlyakov A.N.,
Equationally Noetherian varieties of semigroups and B.Plotkin`s problem}
\thanks{\copyright \ 2023 Shevlyakov A.N.}
\thanks{\rm The work is supported by RSF (grant 22-11-20019)}
\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%


\semrtop \vspace{1cm}
\maketitle {\small

\vspace{-12pt}
\centerline{{\it Communicated by} {\sc P.P. Petrov} ({\it filled by Editorial board})}

\bigskip
\bigskip

\begin{quote}
\noindent{\bf Abstract:} We consider systems of semigroup equations with constants. A semigroup $S$ is called equationally Noetherian if any system of equations is equivalent over $S$ to a finite subsystem. In the current paper we describe all semigroup varieties that consist of equationally Noetherian semigroups. Our result solves the problem of B.Plotkin for semigroup varieties.\medskip

\noindent{\bf Keywords:} semigroups, varieties, universal algebraic geometry.
 \end{quote}
}

\bigskip

\section{Introduction}

The universal algebraic geometry is the subject on the edge of algebra and model theory and it deals with equations over arbitrary algebraic structures. Obviously, algebraic structures of the same variety have common equational properties. It allows us to develop a uniform algebraic geometry for all algebraic structures of a fixed variety. 

An atomic formula of a semigroup language $\LL=\{\cdot\}$ is called an $\LL$-equation (examples of $\LL$-equations are: $x_1x_2=x_2x_1$, $x_1^2=x_1$, $x_1x_2^nx_1=x_2$). A semigroup $S$ is  equationally Noetherian if any system of $\LL$-equations in a finite number of variables is equivalent over $S$ to its finite subsystem. 

The following general problem for  varieties of equationally Noetherian algebraic structures was posed by B.~Plotkin in~\cite{5}.

\bigskip

\noindent {\bf Problem 1.} Describe all varieties $\Vbf$ of algebraic structures such that every $A\in\Vbf$ is equationally Noetherian?

\bigskip

Problem 1 admits positive solution in certain varieties. For example, any abelian variety of groups consists of equationally Noetherian elements (see~\cite{2} for more details). In the class of semigroups, the following varieties 
\begin{itemize}
    \item the variety of semigroups with zero multiplication,
    \item the variety of left (right) zero semigroups
\end{itemize}
satisfy the statement of Problem~1.   

We see above that the language $\LL$ defines the class of equations with no constants. One can reformulate Problem~1 for equations with constants.

\bigskip

\noindent {\bf Problem 2.} Describe all varieties $\Vbf$ of $\LL$-algebras such that each $A\in\Vbf$ is  equationally Noetherian with respect to  equations with constants from $A$ (i.e. $A$ is equationally Noetherian in the language $\LL(A)=\LL\cup \{a\mid a\in A\}$)?

\bigskip

In the current paper we completely solve Problem~2 for semigroup varieties. The description of such varieties $\Vbf$ is given in Theorem~\ref{th:result}.  %Theorem~\ref{th:identities} describes these varieties $\Vbf$ by a list of identities.










\section{Basic notions}


Let $S$ be a semigroup, and $E(S)=\{e\in S\mid ee=e\}$ be the set of its {\it idempotents}. A set $J\subseteq S$ is called an {\it ideal} if $sJ\subseteq J$ and $Js\subseteq J$ for any $s\in S$. A semigroup $S$ is {\it simple} if it has a unique ideal $J=S$. By $\lb a\rb_S$ we denote the ideal of $S$ generated by an element $a\in S$. 

An element $s\in S$ is {\it reducible} if there exist $a,b\in S$ with $s=ab$. The set of all reducible elements of a semigroup $S$ is denoted by $Red(S)$. Obviously, $Red(S)$ is an ideal of $S$, and $E(S)\subseteq Red(S)$.

An idempotent $e$ is {\it primitive} if for any $f\in E(S)\setminus\{0\}$ the equalities $fe=ef=f$ imply $f=e$. A simple semigroup $S$ is {\it completely simple} if it contains a primitive idempotent.

The next classic theorem describes completely simple semigroups (see e.g.~\cite{1}).

\begin{theorem}
\label{th:rees}
For any completely simple semigroup $S$ there exists a group $G$ and sets $I,\Lambda$ such that  $S$ is isomorphic to the set of triples $(\lambda,g,i)$, $g\in G$, $\lambda\in\Lambda$, $i\in I$ with the multiplication
\begin{equation*}
    (\lambda,g,i)(\mu,h,j)=(\lambda,gp_{i\mu}h,j),
    \label{eq:product_in_cs_semigroup}
\end{equation*}
where $p_{i\mu}\in G$ is an element of a matrix $\mathbf{P}$ such that
\begin{enumerate}
\item $\mathbf{P}$ consists of  $|I|$ rows and $|\Lambda|$ columns;
\item the first index of the sets $\Lambda,I$ is denoted by $1$;
\item the matrix $\mathbf{P}$ is $(1,1)$-\textit{normalised}, i.e. 
\[
p_{i1}=p_{1\lambda}=1\in G\mbox{ for all }\lambda\in\Lambda,\;i\in I.
\]
\end{enumerate}
\end{theorem}

Following Theorem~\ref{th:rees}, we denote any completely simple semigroup $S$ by $S=(\Lambda,G,I,\mathbf{P})$. The group $G$ and the matrix $\mathbf{P}$ are called \textit{the structural group} and~\textit{sandwich-matrix}, respectively. If the sandwich-matrix $\mathbf{P}$ consists of $1\in G$ we will use the short denotation $(\Lambda,G,I)$ instead of $(\Lambda,G,I,\mathbf{P})$.   

Suppose a semigroup $S$ has an ideal $J$. Then the factor-semigroup $H=S/J$ is called {\it the Rees factor semigroup}. The semigroup $H$ always contains the zero (the ideal $J$ is mapped into the zero of $H$).

%Let $\lb a\rb\subseteq S$ be the subsemigroup generated by an element $a\in S$. The {\it period} of $a$ is $|\lb a\rb|$. The {\it index} of an element $a$ with a period $n$ is the minimal number $m$ such that $a^m=a^n$ (obviously, $m<n$). It is easy to prove that the subsemigroup $\lb a\rb$ has idempotent $a^{n-m}$ if $a$ has a period $n$ and an index $m$. 

An infinite direct power $S^\infty$ of a semigroup $S$ is the set of all sequences
\[
(a_1,a_2,\ldots,a_n,\ldots),\; a_i\in S,
\]
and with the element-wise multiplication.

Let $\LL=\{\cdot\}$ be the semigroup language. An {\it equation over  $\LL$ ($\LL$-equation)} is an equality of two $\LL$-terms: $\tau(X)=\sigma(X)$. The examples of $\LL$-equations are the following: $xy=yx$, $x^2=x$, $xyx=yxy$. 

A system $\Ss$ of $\LL$-equations ($\LL$-system for shortness) is an arbitrary set of $\LL$-equations. The set of all solutions of $\Ss$ in variables $X=\{x_1,x_2,\ldots,x_n\}$ over a semigroup $S$ is denoted by $\V_S (\Ss(X))\subseteq S^n$. 

A semigroup $S$ is {\it equationally Noetherian} if any infinite $\LL$-system $\Ss$ in variables $X=\{x_1,x_2,\ldots,x_n\}$ is equivalent to a finite subsystem $\Ss^\pr\subseteq \Ss$ over $S$  (the equivalence of $\Ss,\Ss^\pr$ means $\V_S (\Ss^\pr)=\V_S (\Ss)$). 

By $\LL (S)$ we denote the language $\LL\cup\{s\mid s\in S\}$ extended by new constant symbols. The language extension allows us to use constants in equations. The examples of equations in the extended language $\LL (S)$ are the following: $x^2=s$, $xs=sx$, $s_1x=s_2x$, where $s,s_1,s_2\in S$. Obviously, the class of $\LL (S)$-equations is wider than the class of $\LL$-equations, so an equationally Noetherian (in the language $\LL$) semigroup $S$ may lose this property in the language $\LL (S)$. To avoid the ambiguity, we say that $S$ is {\it $S$-equationally Noetherian} if $S$ is equationally Noetherian in the language $\LL (S)$.

Let 
\[
\tau(X)=u_1\ldots u_m
\]
be an $\LL (S)$-term, $u_i\in S\sqcup X$, $X=\{x_1,\ldots,x_n\}$ and $u_{i+1}\notin S$ if $u_i\in S$. Each $u_i$ is said to be a {\it literal} of the term $\tau(X)$. Moreover, the number $m$ is called the {\it length} of $\tau(X)$ and denoted  by $|\tau|$. For example, if $S$ is the free semigroup generated by elements $a,b$ then the length of the $\LL (S)$-term $x_1abx_2x_3b^3$ is $5$. 

Let $S$ be a semigroup, and $\tau(X),\sigma(X)$ be $\LL$-terms. We say that $S$ {\it satisfies the identity}  $\tau(X)=\sigma(X)$ (and denote $S\models [\tau(X)=\sigma(X)]$) if for any $n$-tuple $(s_1,s_2,\ldots,s_n)\in S^n$ it holds $\tau(s_1,s_2,\ldots,s_n)=\sigma(s_1,s_2,\ldots,s_n)$. The class of semigroups $\Vbf=\{S\mid S\models [\tau(X)=\sigma(X)]\mbox{ for all } \tau(X)=\sigma(X)\in \mathcal{I}\}$ is called the {\it variety} defined by a set of identities $\mathcal{I}$. Indeed, any variety is closed under the taking  direct products, homomorphic images and subsemigroups. 

Let $\var(S)$ denote the variety generated by a semigroup $S$, i.e. $\var(S)$ consists of all semigroups which satisfy all identities $\tau(X)=\sigma(X)$ such that $S\models [\tau(X)=\sigma(X)]$. The variety $\var(\mathbf{K})$ generated by a class of semigroups $\mathbf{K}$ consists of all semigroups $S$ such that $S\models [\tau(X)=\sigma(X)]$ for any identity $\tau(X)=\sigma(X)$ with $T\models \tau(X)=\sigma(X)$ for each $T\in \mathbf{K}$.



\section{Main result: structure}

Since any variety is closed under direct powers, one should study semigroups that define equationally Noetherian direct powers.

\begin{lemma}
Let $S$ be a semigroup such that the direct power $S^\infty$ is $S^\infty$-equationally Noetherian. Then the solution set $Y=\V_S (xa=xb)$ (respectively, $Y=\V_S (ax=bx)$) is either $\emptyset$ or $S$.
\label{l:for_semigroups}
\end{lemma}
\begin{proof}
Let us consider an equation $xa=xb$, $a,b\in S$. If $Y\notin\{\emptyset,S\}$, there exists $c,d$ such that $ca=cb$, $da\neq db$. Let us consider the following system of $\LL (S^\infty)$-equations
\[
\Ss=\begin{cases}
\x(a,a,a,\ldots)=\x(a,a,a,\ldots),\\
\x(a,a,a,\ldots)=\x(b,a,a,\ldots),\\
\x(a,a,a,\ldots)=\x(b,b,a,\ldots),\\
\x(a,a,a,\ldots)=\x(b,b,b,\ldots),\\
\ldots
\end{cases}
\]
An element $\x=(x_1,x_2,x_3,\ldots)$ from the solution set of $\Ss$ has $x_i\neq d$ for all $i$. However, the solution set of the first $n$-equations of $\Ss$ contains the point
\[
(\underbrace{c,c,\ldots,c}_{\mbox{$n-1$ times}},d,d,d,\ldots).
\]
Thus, $\Ss$ is not equivalent to its finite subsystems, and $S^\infty$ is not $S^\infty$-equationally Noetherian.
\end{proof}

\begin{corollary}
\label{cor:S_with_zero}
Let $S$ be a semigroup with a zero $0$ such that $S^\infty$ is $S^\infty$-equationally Noetherian. Then $S$ has a zero multiplication.
\end{corollary}
\begin{proof}
Let $a,b\in S$ and consider an equation $ax=0x$. Since the given equation is consistent over $S$ ($0\in\V_S (ax=0x)$), Lemma~\ref{l:for_semigroups} provides that $b\in\V_S (ax=0x)$. Thus, $ab=0b=0$.
\end{proof}

One can consider any group $G$ in the semigroup language $\LL$. For direct powers of groups we have the following result.

\begin{lemma}
For any non-abelian group $G$ the direct power $G^{\infty}$ is not $G^\infty$-equationally Noetherian.
\label{l:for_groups}
\end{lemma}
\begin{proof}
Let us consider the following system of $\LL (G^\infty)$-equations
\[
\Ss=\begin{cases}
\x(1,1,1,\ldots)=(1,1,1,\ldots)\x,\\
\x(a,1,1,\ldots)=(a,1,1,\ldots)\x,\\
\x(a,a,1,\ldots)=(a,a,1,\ldots)\x,\\
\x(a,a,a,\ldots)=(a,a,a,\ldots)\x,\\
\ldots
\end{cases}
\]
where an element $a\in G$ does not commute with some $b\in G$.

An element $\x=(x_1,x_2,x_3,\ldots)$ from the solution set of $\Ss$ has $x_i\neq b$ for all $i$. However, the solution set of the first $n$-equations of $\Ss$ contains the point
\[
(\underbrace{1,1,\ldots,1}_{\mbox{$n-1$ times}},b,b,b,\ldots).
\]
Thus, $G^\infty$ is not $G^\infty$-equationally Noetherian.
\end{proof}

On the other hand, every abelian group $G$ is $G$-equationally Noetherian (see~\cite{2}).







In Lemmas~\ref{l:ideal_is_more_red}--\ref{l:aeb=afb} we assume that a semigroup variety $\Vbf$ satisfies the following statement:
\begin{equation*}
    \mbox{``every }S\in \Vbf\mbox{ is $S$-equationally Noetherian''}.
    \label{eq:Vbf_statement}
\end{equation*}





\begin{lemma}
Let $I$ be an ideal of $S\in\Vbf$. Then $Red(S)\subseteq I$.
\label{l:ideal_is_more_red}
\end{lemma}
\begin{proof}
The Rees factor semigroup $S/I$ belongs to $\Vbf$ and contains a zero. By Corollary~\ref{cor:S_with_zero}, $S/I$ has zero multiplication. Hence for all $a,b\in S$ it holds $ab\in I$. Thus, $Red(S)\subseteq I$. 
\end{proof}





\begin{lemma}
The set of all reducible elements $Red(S)$ of a semigroup $S\in\Vbf$ is a simple semigroup.
\label{l:red_is_simple}
\end{lemma}
\begin{proof}
Assume there exists an element $x\in R=Red(S)$ such that $\lb x\rb_R\subset R$. Consider an ideal 
\[
I=\lb x^5\rb_S=\{axxxxxb\mid a,b\in S^1\}
\]
of the semigroup $S$. By Lemma~\ref{l:ideal_is_more_red}, we obtain the contradiction:
\begin{eqnarray*}
R&\subseteq& I=\{axxxxxb\mid a,b\in S^1\}\\
&=&\{r_1xr_2\mid r_1\in S^1xx\subseteq R, r_2\in xxS^1\subseteq R\}\\
&\subseteq& \{r_1xr_2\mid r_1,r_2\in R\}\\
&=&\lb x\rb_R.
\end{eqnarray*}
\end{proof}



\begin{lemma}
\label{l:finite_orders_of_elements}
Any element of a semigroup $S\in \Vbf$ has finite order. 
\end{lemma}
\begin{proof}
Assume that $s\in S$ generates an infinite cyclic semigroup $C_\infty$. Further in this lemma we consider the semigroup $C_\infty$ in the additive form with the natural linear order $\leq$.

Then $C_\infty\oplus C_\infty\in\Vbf$. Let us show that the two-element idempotent semigroup $L_2=\{0,1\}$ is a factor of a subsemigroup of $C_\infty\times C_\infty$. Indeed, one can decompose the subsemigroup $\{(a,b)\mid a\leq b\}\subseteq C_\infty\times C_\infty$ into the union of the equivalence classes $A_1=\{(a,a)\mid a\in C_\infty\}$, $A_0=\{(a,b)\mid a<b, a,b\in C_\infty\}$.

The multiplication over $A_1,A_0$ is given by $A_iA_i=A_i, A_0A_1=A_1A_0=A_0$, and therefore $\{A_1,A_0\}$ is isomorphic to $L_2$. Thus, $L_2\in\Vbf$ and $L_2^\infty\in\Vbf$, but $L_2^\infty$ is not $L_2^\infty$-equationally Noetherian (Corollary~\ref{cor:S_with_zero}). 
\end{proof}

Let us give some facts about the bicyclic semigroup $B=\lb a,b\mid ab=1\rb$

\begin{proposition}(\textup{\cite{1}}, Theorem 2.54)
The bicyclic semigroup $B$ is embedded into a simple but not completely simple semigroup $S$ if $S$ has at least one idempotent.
\label{pr:Andersen}
\end{proposition}



\begin{proposition}(\textup{\cite{3}}, Theorem 4.1)
The bicyclic semigroup $B$ is not $B$-equationally Noetherian.
\label{pr:Plandowsky}
\end{proposition}

\begin{lemma}
\label{l:Red_is_completely_simple}
Let $S\in \Vbf$ then $R=Red(S)$ is a completely simple semigroup.
\end{lemma}
\begin{proof}
By Lemma~\ref{l:red_is_simple}, $R$ is simple. By Lemma~\ref{l:finite_orders_of_elements}, any element of $R\in\Vbf$ has a finite order. Hence, $R$ contains idempotents. 

If $R$ is not completely simple, it contains the bicyclic semigroup $B$ (Proposition~\ref{pr:Andersen}), and therefore $B\in\Vbf$. Since $B$ is not equationally Noetherian (Proposition~\ref{pr:Plandowsky}), we obtain a contradiction with the choice of $\Vbf$. 
\end{proof}

Lemma~\ref{l:Red_is_completely_simple} provides that $Red(S)$ has the form $(\Lambda,G,I,\mathbf{P})$.

\begin{lemma}
\label{l:structural_group_is_abelian}
For a variety $\Vbf$ there exists a number $n\geq 1$ such that any group $G\in\Vbf$ is abelian and of the period $n$ (i.e. $g^n=1$ for all $g\in G$).
\end{lemma}
\begin{proof}
By Lemma~\ref{l:for_groups}, any group $G$ is abelian. Assume that for any number $n$ there exists a group $G\in \Vbf$ and $g\in G$ with $g^n\neq 1$. 

Let $\mathbf{G}\subseteq \Vbf$ be all groups from the variety $\Vbf$. The class $\mathbf{G}$ is not empty, since any semigroup from $\Vbf$ contains a subgroup (Lemma~\ref{l:Red_is_completely_simple}).

Let us consider the semigroup variety $\var(\mathbf{G})$ and take an arbitrary identity $\tau(X)=\sigma(X)$ which holds in $\var(\mathbf{G})$. 

First, we prove that the identity $\tau(X)=\sigma(X)$ is {\it balanced}, i.e. for any letter $x_i\in X$ the number of occurrences of $x_i$ in $\tau(X)$  coincides with the number of occurrences in $\sigma(X)$.

Assume the converse: the number of occurrences of $x_i$ in $\tau(X)$ (respectively, $\sigma(X)$) equals $k$ (respectively, $l$) and $k\neq l$. 
If we assign $x_j=1$ for all $j\neq i$ we obtain the identity $x_i^k=x_i^l$. Equivalently, the identity $x_i^{k-l}=1$ holds for any group $G\in \var(\mathbf{G})$. It follows that any group from $\Vbf$ has the period $k-l$.

Thus, all identities of $\var(\mathbf{G})$ are balanced. Hence, the infinite cyclic group $Z$ belongs to this variety. The last fact contradicts Lemma~\ref{l:finite_orders_of_elements}, since all nontrivial elements of $Z$ have infinite order.
\end{proof}


\begin{lemma}
\label{l:sandwich_matrix_is_1}
Let $S\in\Vbf$ then $Red(S)$ is isomorphic to $(\Lambda,G,I,\mathbf{P})$ with $p_{i\lambda}=1$ for all $i\in I$, $\lambda\in \Lambda$.
\end{lemma}
\begin{proof}
Recall that the sandwich-matrix $\mathbf{P}$ is $(1,1)$-normalized. Let us consider an equation:
\[
x(\lambda,1,1)=x(1,1,1).
\]
Obviously, this equation satisfies the point $x=(1,1,1)$. According to Lemma~\ref{l:for_semigroups}, this equation should be satisfied by any $x\in S$. In particular, we have 
\[
(1,p_{i\lambda},1)=(1,1,i)(\lambda,1,1)=(1,1,i)(1,1,1)=(1,1,1).
\]
Thus, $p_{i\lambda}=1$.
\end{proof}



\begin{lemma}
\label{l:multiplication_in_S}
Let $S\in\Vbf$ and $R=Red(S)=(\Lambda,G,I)$. Then for the any $a\in S$ there exist $g_a\in G$, $\lambda_a\in \Lambda$, $i_a\in I$ such that the multiplication in $S$ is defined as follows:
\begin{equation}
\label{eq:a(mu,g,i)}
a(\mu,g,i)=(\lambda_a,g_ag,i), \; (\mu,g,i)a=(\mu,gg_a,i_a) 
\end{equation}
for all $\mu\in\Lambda$, $g\in G$, $i\in I$.
\end{lemma}
\begin{proof}
If $a=(\lambda_a,g_a,i_a)\in R$, we immediately obtain~(\ref{eq:a(mu,g,i)}). Assume below $a\in S\setminus R$.

Since $a(1,1,1)\in R$, there exists $\lambda_a,g_a,i^\pr_a$ such that $a(1,1,1)=(\lambda_a,g_a,i^\pr_a)$. We have:
\[
(\lambda_a,g_a,1)=(\lambda_a,g_a,i^\pr_a)(1,1,1)=a(1,1,1)(1,1,1)=a(1,1,1)=(\lambda_a,g_a,i^\pr_a),
\]
hence $i^\pr_a=1$.

Since the equation $x(\mu,1,1)=x(1,1,1)$ has a solution $(1,1,1)$, Lemma~\ref{l:for_semigroups} provides 
\begin{equation}
a(\mu,1,1)=a(1,1,1)=(\lambda_a,g_a,1).
\label{eq:lllll}    
\end{equation}
Therefore, 
\[
a(\mu,g,i)=a(\mu,1,1)(1,g,i)=(\lambda_a,g_a,1)(1,g,i)=(\lambda_a,g_ag,i)
\]
for all $\mu\in\Lambda$, $g\in G$, $i\in I$.
Similarly, one can prove 
\[
(\mu,g,i)a=(\mu,gg_a^\pr,i_a)
\]
for all $\mu,g,i$. Let us prove $g_a^\pr=g_a$. Namely,
\[
(1,g_a^\pr,1)=(1,g_a^\pr,i_a)(1,1,1)=(1,1,1)a(1,1,1)=(1,1,1)(\lambda_a,g_a,1)=(1,g_a,1).
\]
Thus, we proved~(\ref{eq:a(mu,g,i)}).
\end{proof}


\begin{lemma}
\label{l:multiplication_in_S_2}
Let $S\in\Vbf$, and $Red(S)=(\Lambda,G,I)$, then
\begin{equation}
\label{eq:ab}
ab=(\lambda_a,g_ag_b,i_b) 
\end{equation}
for any $a,b\in S$.
\end{lemma} 
\begin{proof}
The product $ab$ belongs to $R=Red(S)$, so $ab=(\mu,g,i)$ for some $\mu\in\Lambda$, $g\in G$, $i\in I$.

Using~(\ref{eq:a(mu,g,i)}), we have
\[
(1,g_ag_b,i_b)=(1,g_a,i_a)b=(1,1,1)ab=(1,1,1)(\mu,g,i)=(1,g,i),
\]
hence $g=g_ag_b$, $i=i_b$. On the other hand,
\[
(\mu,g_ag_b,1)=(\mu,g_ag_b,i_b)(1,1,1)=ab(1,1,1)=a(\lambda_b,g_b,1)=(\lambda_a,g_ag_b,1),
\]
Thus, $\mu=\lambda_a$, and~(\ref{eq:ab}) holds.
\end{proof}





\begin{lemma}
\label{l:aeb=afb}
Any $S\in\Vbf$ satisfies 
\begin{equation}
    aeb=afb\mbox{ \textup{for all} }a,b\in S,\; e,f\in E(S).
    \label{eq:aeb=afb}
\end{equation}
\end{lemma} 
\begin{proof}
Since any idempotent belongs to $Red(S)$, and any entry of the sandwich-matrix of $Red(S)$ equals $1$ (Lemma~\ref{l:sandwich_matrix_is_1}), arbitrary idempotents of $S$ have the form: $e=(\lambda_e,1,i_e)$, $f=(\lambda_f,1,i_f)$ for some $\lambda_e,\lambda_f\in\Lambda$, $i_e,i_f\in I$. Using~(\ref{eq:ab}), we have
\[
aeb=(\lambda_a,g_a,i_e)b=(\lambda_a,g_ag_b,i_b)=(\lambda_a,g_a,i_f)b=afb.
\]
\end{proof}

\begin{lemma}
\label{l:aeb=afb_implies}
Suppose a semigroup $S$ satisfies~(\ref{eq:aeb=afb}), and $Red(S)=(\Lambda,G,I,\mathbf{P})$ is completely simple. Then  
\begin{enumerate}
    \item any entry of $P$ equals $1$,
    \item for any $a,b\in S$ the equality~(\ref{eq:ab}) holds.
\end{enumerate}
\end{lemma}
\begin{proof}
Let $e,f\in E(S)\subseteq Red(S)$. The equality~(\ref{eq:aeb=afb}) gives $fef=fff=f$. Let $f=(1,1,1)$ and $e$ has the indexes $\lambda\in \Lambda$, $i\in I$. According to the multiplication if $Red(S)$, $e$ is the triple $e=(\lambda,p_{i\lambda}^{-1},i)$. The equality $fef=f$ gives
\[
(1,1,1)(\lambda,p_{i\lambda}^{-1},i)(1,1,1)=(1,1,1)\Rightarrow (1,p_{i\lambda}^{-1},1)=(1,1,1)\Rightarrow p_{i\lambda}=1.
\]

Let us prove~(\ref{eq:ab}). If $e=(\lambda,1,i)$, $f=(\mu,1,j)$, $b=(1,1,1)$ then~(\ref{eq:aeb=afb}) gives
\[
a(\lambda,1,1)=a(\mu,1,1). 
\]
Hence, the exists $\lambda_a,g_a,i^\pr_a$ with $a(\lambda,1,1)=(\lambda_a,g_a,i^\pr_a)$ for any $\lambda\in\Lambda$. The further proof coincides with Lemmas~\ref{l:multiplication_in_S} and~\ref{l:multiplication_in_S_2}.
\end{proof}






Suppose a semigroup satisfies~(\ref{eq:ab}) and $Red(S)$ is completely simple. One can define  a mapping:
\[
\phi\colon S\to Red(S),\; \phi(s)=(\lambda_s,g_s,i_s).
\]
Moreover, the equality~(\ref{eq:ab}) implies that $\phi$ is a homomorphism of semigroups.

Let $P=(p_1,\ldots,p_n)\in S^n$, then $\phi(P)$ denotes the point $(\phi(p_1),\ldots,\phi(p_n))\in Red(S)^n$.

Let $\tau(X)=u_1\ldots u_m$ be an $\LL(S)$-term with literals $u_i\in S\sqcup X$. One can define the $\LL(R)$-term $\tau_\phi(X)=(u_1)_\phi\ldots (u_m)_\phi$ as follows: 
\[
(u_i)_\phi=\begin{cases}
\phi(u_i) \mbox{ if } u_i\in S,\\
u_i\mbox{ if } u_i\in X
\end{cases}
\]

For example, if $\tau(X)=x_1ax_2b$ then $\tau_\phi(X)=x_1(\lambda_a,g_a,i_a)x_2(\lambda_b,g_b,i_b)$.

\begin{lemma}
Let $\tau(X)$ be an $\LL (S)$-term with $|\tau|\geq 2$ and a semigroup $S$ satisfies~(\ref{eq:ab}). Then for any $P\in S^n$ we have the equalities:
\begin{equation}
    \tau(P)=\tau(\phi(P))=\tau_\phi(P)=\tau_\phi(\phi(P)).
    \label{eq:4_equalities}
\end{equation}
\end{lemma}
\begin{proof}
It is sufficient to prove the equalities for any term of length 2 (the proof for longer terms can be done by the induction). Thus, we have the following types of $\LL (S)$-terms of length 2 (here $x,y\in X$, $s\in S$):
\begin{enumerate}
    \item $\tau(X)=xy$,
    \item $\tau(X)=xs$,
    \item $\tau(X)=sx$.
\end{enumerate}
Let us consider the first case. For a point $P=(a,b)$ we have $\tau_\phi(X)=xy$, $\phi(P)=((\lambda_a,g_a,i_a),(\lambda_b,g_b,i_b))$. One can directly prove that 
\[
\tau(P)=\tau(\phi(P))=\tau_\phi(P)=\tau_\phi(\phi(P))=(\lambda_a,g_ag_b,i_b),
\]
and~(\ref{eq:4_equalities}) holds.

For the second case we have (here $P=a$): $\phi(P)=(\lambda_a,g_a,i_a)$, $\tau_\phi(X)=x\sigma_\phi=x(\lambda_s,g_s,i_s)$, and 
\[
\tau(P)=\tau(\phi(P))=\tau_\phi(P)=\tau_\phi(\phi(P))=(\lambda_a,g_ag_s,i_s).
\]

The third case is similar to the previous one.
\end{proof}




In~\cite{4} it was proved the following criterion of the Noetherian property for completely simple semigroups.

\begin{proposition}\textup{(\cite{4}, Theorem 8)}
\label{pr:Neotherian_cs_semigroups}
Let $S=(\Lambda,G,I,\mathbf{P})$ be a completely simple semigroup. The semigroup is $S$-equationally Noetherian iff $G$ is $G$-equationally Noetherian.
\end{proposition}

In particular, if $G$ is abelian, any semigroup $S=(\Lambda,G,I,\mathbf{P})$ is $S$-equationally Noetherian.



\begin{theorem}
A variety $\Vbf$ completely consists of $S$-equationally Noetherian semigroups iff there exists a number $n$ such that every $S\in\Vbf$ satisfies the following conditions:
\begin{enumerate}
    \item the set of reducible elements $R=Red(S)$ is isomorphic to a completely simple semigroup $(\Lambda,G,I,\mathbf{P})$;
    \item the group $G$ is abelian and of the period $n$;
    \item $S$ satisfies~(\ref{eq:aeb=afb}).
\end{enumerate}
\label{th:result}
\end{theorem}
\begin{proof}
The ``only if'' part of the theorem directly follows from Lemmas~\ref{l:Red_is_completely_simple},~\ref{l:structural_group_is_abelian},~\ref{l:aeb=afb}.

Let us prove the ``if'' part of the theorem. By Lemma~\ref{l:aeb=afb_implies}, we obtain that any entry of the sandwich-matrix $\mathbf{P}$ equals $1$ and~(\ref{eq:ab}) holds.

Let us consider a system of $\LL (S)$-equations $\Ss=\{t_j(X)=s_j(X)\mid j\in J\}$ over a semigroup $S$ in variables $X=\{x_1,\ldots,x_n\}$. 

Applying the map $\phi\colon S\to R$, we obtain the system  $\phi(\Ss)=\{(t_j)_\phi(X)=(s_j)_\phi(X)\mid j\in J\}$ of $\LL (R)$-equations. By Proposition~\ref{pr:Neotherian_cs_semigroups}, the system $\phi(\Ss)$ is equivalent to a finite subsystem $\Ss_\phi\subseteq\phi(\Ss)$ over the semigroup $R$.

For every equation from $\Ss_\phi$ we choose a pre-image in $\Ss$ to collect a finite subsystem $\Ss^\pr\subseteq \Ss$. Let 
\[
\Ss^1=\{\tau(X)=\sigma(X)\mid |\tau(X)|=1\mbox{ or } |\sigma(X)|=1\}\subseteq \Ss.
\]
Obviously, $\Ss^1$ is a union
\[
\Ss^1=\Ss^{11}\cup\bigcup_{i=1}^n\Ss_i^1,
\]
where 
\begin{enumerate}
    \item $\Ss^{11}\subseteq \Ss$ is the set of all equations $\pi(X)=\rho(X)\in\Ss$ with $|\pi|=|\rho|=1$;
    \item $\Ss_i^1\subseteq \Ss$ is the set of all equations $\pi(X)=\rho(X)\in\Ss\setminus\Ss^{11}$ such that either $\pi(X)$ or $\rho(X)$ equals $x_i\in X$.
\end{enumerate}

%\; \Ss_i^1=\bigcup_s\{x_i=\sigma(X)\}\cup\bigcup_t\{\tau(X)=x_i\}, \; \Ss^{11}=\{\tau(X)=\sigma(X)\mid |\tau|=|\sigma|=1\}.
%\]
Let us choose an arbitrary equation $\mathbf{Eq}_i$ from each system $\Ss^1_i$ and denote 
\[
\Ss^{\pr\pr}=\Ss^\pr\cup\bigcup_i \mathbf{Eq}_i\cup \Ss^{11}\subseteq \Ss.
\]
Let us prove that the finite subsystem $\Ss^{\pr\pr}$ is equivalent to $\Ss$ over $S$.

Assume there exists a point $P=(p_1,\ldots,p_n)$ with $P\in\V_S (\Ss^{\pr\pr})\setminus\V_S (\Ss)$. In other words, there exists an equation $\tau(X)=\sigma(X)\in \Ss\setminus\Ss^{\pr\pr}$ and $\tau(P)\neq \sigma(P)$.

Since $\phi$ is a homomorphism, we have $\phi(P)\in\V_R(\phi(\Ss^{\pr\pr}))$. By the choice of the system $\Ss^\pr$, we have $\phi(P)\in\V_R(\phi(\Ss))$, in particular
\begin{equation}
    \phi(P)\in\V_R(\tau_\phi(X)=\sigma_\phi(X)).
    \label{eq:222222}
\end{equation}


By~(\ref{eq:4_equalities}), $P\in\V_S (\tau(X)=\sigma(X))$ if $|\t|,|\s|\geq 2$. Since $\tau(P)\neq\sigma(P)$, at least one length of terms $\t,\s$ equals $1$. The case $|\t|=|\s|=1$ is impossible, since the system $\Ss^{11}$ is included into $\Ss^{\pr\pr}$.

Thus, we may assume that $\tau(X)=x_i$ and $|\s|\geq 2$. The choice of the equation  $x_i=\sigma(X)$ gives $p_i\neq \sigma(P)$. By~(\ref{eq:4_equalities}),~(\ref{eq:222222}), $\phi(p_i)=\sigma_\phi(\phi(P))=\sigma(P)$.

Hence, $p_i\neq \phi(p_i)$ and the definition of $\phi$ gives  $p_i\in S\setminus R$. However, the system $\Ss^{\pr\pr}$ contains an equation of the form $x_i=\sigma(X)$, $|\sigma|\geq 2$ with $p_i=\sigma(P)$. Since $|\sigma|\geq 2$, we have $p_i=\sigma(P)\in R$, a contradiction.
\end{proof}

\begin{example}
Let $\Vbf$ be the variety of rectangular bands, i.e. $\Vbf$ is defined by the identities $xx=x$, $xyz=xz$. It is well-known fact that any semigroup from $\Vbf$ is isomorphic to $(\{1\},\Lambda,I)$ for some index sets $\Lambda$, $I$. 

Let us check the conditions of Theorem~\ref{th:result} for $\Vbf$. Namely, for $S\in\Vbf$ we have $S=Red(S)=(\Lambda,\{1\},I)$, and $n=1$. The property~(\ref{eq:aeb=afb}) follows from the identity $xyz=xz$. 

Thus, $\Vbf$ completely consists of equationally Noetherian semigroups.
\end{example}






























\bigskip

\end{document}
